Author's response to reviews

Title: Family physician and endocrinologist coordination as the basis for diabetes care in clinical practice

Authors:

Alejandra Duran (aduranh@hotmail.com)
Isabelle Runkle (irunkle.hcsc@salud.madrid.org)
Pilar Matia (mmatia.hcsc@salud.madrid.org)
Paz de Miguel (mdemiguel.hcsc@salud.madrid.org)
Sofia Garrido (sgarrido.gapm07@salud.madrid.org)
Emilio Cervera (ecervra.gapm07@salud.madrid.org)
Dolores Fernandez (dfernandez.hcsc@salud.madrid.org)
Pilar Torres (mторres.hcsc@salud.madrid.org)
Tomas Lillo (tillio.hcsc@salud.madrid.org)
Patricia Martin (pmartinojarmarcos@yahoo.es)
Lucio Carerizo (lcabrerizo.hcsc@salud.madrid.org)
Nuria Garcia de la Torre (nrialobo@hotmail.com)
Jose R Calle (jrcalle@telefonica.net)
Jose Ibarra (jibarra@hotmail.com)
Aniceto L Charro (acharro.hcsc@salud.madrid.org)
Alfonso L Calle-Pascual (acalle.hcsc@salud.madrid.org)

Version: 7 Date: 7 March 2008

Author's response to reviews: see over
Dear Sir,
I send you the revised form of the m.n FAMILY PHYSICIAN AND ENDOCRINOLOGIST COORDINATION……., to BMC Endocrine Disorder. Thank you very much for their appreciate comments.
The paper has been reedited and grammatical errors corrected. According to reviewers comments we insert in the m.s. some paragraph, reedited abstract, introduction and discussion section and table 2 as table 3.
We have included an estimate of the cost caused by the hours of teaching programme, as well as we have included the number of sessions and meeting. We have included the estimate of sample size. We have enlarged the reference to the metaanalysis of Griffin. The differences among groups are only different when changes in BMI are valued, and not when the absolute values are compared. We introduce in tables these differences. In our opinion the finding of the paper as presented are of practical relevance. We think that the m.s. is suitable to be published in BMC Endocrine Dis.
Sincerely
Alfonso L. Calle-Pascual, MDPhD